
% ---

\documentclass[12pt,twoside,fleqn]{article}

\usepackage{a4wide,cite}
\usepackage{amsmath,amssymb}
\usepackage{epsfig,rotating}
\usepackage{axodraw}

\setlength{\topmargin}{-.7cm}
\setlength{\textheight}{24cm}
\setlength{\textwidth}{16.5cm}
\setlength{\oddsidemargin}{-.3cm}
\setlength{\evensidemargin}{-.3cm}
\setlength{\headsep}{5mm}
\setlength{\parskip}{5pt}
\setlength{\mathindent}{.5cm}

\numberwithin{equation}{section}
\allowdisplaybreaks[4]

\def \be{\begin{equation}}
\def \ee{\end{equation}}
\newcommand{\bea}{\begin{eqnarray}}
\newcommand{\eea}{\end{eqnarray}}
\def \nn{\nonumber}

\newcommand{\MS}{\ensuremath{\overline{\text{MS}}}}

% ---

\begin{document}
\begin{thebibliography}{99}

\bibitem{Misiak:2006zs}
  M.~Misiak {\it et al.}, Phys.\ Rev.\ Lett. {\bf 98} (2007) 022002 [arXiv:hep-ph/0609232].

\bibitem{Misiak:2004ew}
  M.~Misiak and M.~Steinhauser, Nucl.\ Phys.\ B {\bf 683} (2004) 277 [arXiv:hep-ph/0401041].

\bibitem{Gorbahn:2004my}
  M.~Gorbahn and U.~Haisch, Nucl.\ Phys.\ B {\bf 713} (2005) 291 [arXiv:hep-ph/0411071].

\bibitem{Gorbahn:2005sa}
  M.~Gorbahn, U.~Haisch and M.~Misiak, Phys.\ Rev.\ Lett.\ {\bf 95} (2005) 102004 [arXiv:hep-ph/0504194].

\bibitem{Czakon:2006ss}
  M.~Czakon, U.~Haisch and M.~Misiak, JHEP {\bf 0703} (2007) 008 [arXiv:hep-ph/0612329].

\bibitem{Blokland:2005uk}
  I.~Blokland, A.~Czarnecki, M.~Misiak, M.~Slusarczyk and F.~Tkachov,
  Phys.\ Rev.\ D {\bf 72} (2005) 033014 [arXiv:hep-ph/0506055].

\bibitem{Melnikov:2005bx}
  K.~Melnikov and A.~Mitov, Phys.\ Lett.\ B {\bf 620} (2005) 69 [arXiv:hep-ph/0505097].

\bibitem{Asatrian:2006ph}
  H.~M.~Asatrian, A.~Hovhannisyan, V.~Poghosyan, T.~Ewerth, C.~Greub and T.~Hurth,
  Nucl.\ Phys.\ B {\bf 749} (2006) 325 [arXiv:hep-ph/0605009].
  %%CITATION = NUPHA,B749,325;%%

\bibitem{Asatrian:2006sm}
  H.~M.~Asatrian, T.~Ewerth, A.~Ferroglia, P.~Gambino and C.~Greub,
  Nucl.\ Phys.\ B {\bf 762} (2007) 212 [arXiv:hep-ph/0607316].
  %%CITATION = NUPHA,B762,212;%%

\bibitem{Bieri:2003ue}
  K.~Bieri, C.~Greub and M.~Steinhauser, Phys.\ Rev.\ D {\bf 67} (2003) 114019 [arXiv:hep-ph/0302051].
  %%CITATION = PHRVA,D67,114019;%%

\bibitem{Misiak:2006ab}
  M.~Misiak and M.~Steinhauser, Nucl.\ Phys.\ B {\bf 764} (2007) 62 [arXiv:hep-ph/0609241].

\bibitem{Misiak:2010sk}
  M.~Misiak and M.~Steinhauser, arXiv:1005.1173.
  %%CITATION = ARXIV:1005.1173;%%

\bibitem{Barberio:2008fa}
  E.~Barberio {\it et al.} [Heavy Flavor Averaging Group], arXiv:0808.1297, and on-line update
  at {\sf http://www.slac.stanford.edu/xorg/hfag/rare/winter10/radll/btosg.pdf}
  %%CITATION = ARXIV:0808.1297;%%

\bibitem{Artuso:2009jw}
  M.~Artuso, E.~Barberio and S.~Stone, PMC Phys.\ A {\bf 3} (2009) 3 [arXiv:0902.3743].
  %%CITATION = PMCPA,A3,3;%%

\bibitem{Chen:2001fja}
  S.~Chen {\it et al.} [CLEO Collab.],
  Phys.\ Rev.\ Lett.\ {\bf 87} (2001) 251807 [arXiv:hep-ex/0108032].
  %%CITATION = PRLTA,87,251807;%%

\bibitem{Aubert:2007my}
  B.~Aubert {\it et al.} [BaBar Collab.],
  Phys.\ Rev.\ D {\bf 77} (2008) 051103 [arXiv:0711.4889].
  %%CITATION = PHRVA,D77,051103;%%

\bibitem{Limosani:2009qg}
  A.~Limosani {\it et al.} [Belle Collab.],
  Phys.\ Rev.\ Lett.\  {\bf 103} (2009) 241801 [arXiv:0907.1384].
  %%CITATION = PRLTA,103,241801;%%

\bibitem{Falk:1993dh}
  A.~F.~Falk, M.~E.~Luke and M.~J.~Savage, Phys.\ Rev.\ D {\bf 49} (1994) 3367 [arXiv:hep-ph/9308288].
  %%CITATION = PHRVA,D49,3367;%%

\bibitem{Bigi:1992ne}
  I.~I.~Y.~Bigi, B.~Blok, M.~A.~Shifman, N.~G.~Uraltsev and A.~I.~Vainshtein, arXiv:hep-ph/9212227.
  %%CITATION = HEP-PH/9212227;%%

\bibitem{Bauer:1997fe}
  C.~W.~Bauer, Phys.\ Rev.\ D {\bf 57} (1998) 5611
  [Erratum-ibid.\ D {\bf 60} (1999) 099907] [arXiv:hep-ph/9710513].
  %%CITATION = PHRVA,D57,5611;%%

\bibitem{Gremm:1996df}
  M.~Gremm and A.~Kapustin, Phys.\ Rev.\ D {\bf 55} (1997) 6924 [arXiv:hep-ph/9603448].
  %%CITATION = PHRVA,D55,6924;%%

\bibitem{Ewerth:2009yr}
  T.~Ewerth, P.~Gambino and S.~Nandi, Nucl.\ Phys.\  B {\bf 830} (2010) 278 [arXiv:0911.2175].
  %%CITATION = NUPHA,B830,278;%%

\bibitem{Lee:2006wn}
  S.~J.~Lee, M.~Neubert and G.~Paz, Phys.\ Rev.\  D {\bf 75} (2007) 114005 [arXiv:hep-ph/0609224].
  %%CITATION = PHRVA,D75,114005;%%

\bibitem{Benzke:2010js}
  M.~Benzke, S.~J.~Lee, M.~Neubert and G.~Paz, arXiv:1003.5012.
  %%CITATION = ARXIV:1003.5012;%%

\bibitem{Domingo:2007dx}
  See, e.g., F.~Domingo and U.~Ellwanger, JHEP {\bf 0712} (2007) 090 [arXiv:0710.3714].
  %%CITATION = JHEPA,0712,090;%%

\bibitem{Asatrian:2006rq}
  H.~M.~Asatrian, T.~Ewerth, H.~Gabrielyan and C.~Greub,
  Phys.\ Lett.\ B {\bf 647} (2007) 173 [arXiv:hep-ph/0611123].
  %%CITATION = PHLTA,B647,173;%%

\bibitem{Boughezal:2007ny}
  R.~Boughezal, M.~Czakon and T.~Schutzmeier, JHEP {\bf 0709} (2007) 072 [arXiv:0707.3090].
  %%CITATION = JHEPA,0709,072;%%

\bibitem{Ewerth:2008nv}
  T.~Ewerth, Phys.\ Lett.\  B {\bf 669} (2008) 167 [arXiv:0805.3911].
  %%CITATION = PHLTA,B669,167;%%

\bibitem{Pak:2008qt}
  A.~Pak and A.~Czarnecki, Phys.\ Rev.\ Lett.\ {\bf 100} (2008) 241807 [arXiv:0803.0960].
  %%CITATION = PRLTA,100,241807;%%

\bibitem{Ligeti:1999ea}
  Z.~Ligeti, M.~E.~Luke, A.~V.~Manohar and M.~B.~Wise,
  Phys.\ Rev.\ D {\bf 60} (1999) 034019 [arXiv:hep-ph/9903305].
  %%CITATION = PHRVA,D60,034019;%%

\bibitem{schutzmeier}
  R.~Boughezal, M.~Czakon and T.~Schutzmeier, in preparation;
  M.~Czakon, T.~Huber and T.~Schutzmeier, in preparation.

\bibitem{Misiak:2008ss}
  M.~Misiak, arXiv:0808.3134.
  %%CITATION = ARXIV:0808.3134;%%

\bibitem{Ferroglia:2008yg}
  A.~Ferroglia, Mod.\ Phys.\ Lett.\  A {\bf 23} (2008) 3123 [arXiv:0812.0082].
  %%CITATION = MPLAE,A23,3123;%%

\bibitem{Ewerth:2009yk}
  T.~Ewerth, arXiv:0909.5027.
  %%CITATION = ARXIV:0909.5027;%%

\bibitem{Misiak:2009nr}
  M.~Misiak, Acta Phys.\ Polon.\  B {\bf 40} (2009) 2987 [arXiv:0911.1651].
  %%CITATION = APPOA,B40,2987;%%

\bibitem{Neubert:2004dd}
  M.~Neubert, Eur.\ Phys.\ J.\  C {\bf 40} (2005) 165 [arXiv:hep-ph/0408179].
  %%CITATION = EPHJA,C40,165;%%

\bibitem{cut1}
  R.~E.~Cutkosky, J.\ Math.\ Phys.\  {\bf 1} (1960) 429.
  %%CITATION = JMAPA,1,429;%%

\bibitem{cut2}
  M.~J.~G.~Veltman, Physica {\bf 29} (1963) 186.
  %%CITATION = PHYSA,29,186;%%

\bibitem{cut3}
  E.~Remiddi, Helv.\ Phys.\ Acta {\bf 54} (1982) 364.
  %%CITATION = HPACA,54,364;%%

\bibitem{Anastasiou:2003ds}
  C.~Anastasiou, L.~J.~Dixon, K.~Melnikov and F.~Petriello,
  Phys.\ Rev.\  D {\bf 69} (2004) 094008 [arXiv:hep-ph/0312266].
  %%CITATION = PHRVA,D69,094008;%%

\bibitem{Anastasiou:2002yz}
  C.~Anastasiou and K.~Melnikov, Nucl.\ Phys.\  B {\bf 646} (2002) 220  [arXiv:hep-ph/0207004].
  %%CITATION = NUPHA,B646,220;%%

\bibitem{Anastasiou:2003gr}
  C.~Anastasiou, K.~Melnikov and F.~Petriello, Phys.\ Rev.\  D {\bf 69} (2004) 076010 [arXiv:hep-ph/0311311].
  %%CITATION = PHRVA,D69,076010;%%

\bibitem{Anastasiou:2004vj}
  C.~Anastasiou and A.~Lazopoulos, JHEP {\bf 0407} (2004) 046 [arXiv:hep-ph/0404258].
  %%CITATION = JHEPA,0407,046;%%

\bibitem{diffeq}
%\bibitem{Remiddi:1997ny}
  E.~Remiddi, Nuovo Cim.\  A {\bf 110}, (1997) 1435 [arXiv:hep-th/9711188].
  %%CITATION = NUCIA,A110,1435;%%

\bibitem{Argeri}
  M.~Argeri and P.~Mastrolia,
  %``Feynman Diagrams and Differential Equations,''
  Int.\ J.\ Mod.\ Phys.\  A {\bf 22} (2007) 4375
  [arXiv:0707.4037].
  %%CITATION = IMPAE,A22,4375;%%

\bibitem{Binoth:2000ps}
  T.~Binoth and G.~Heinrich, Nucl.\ Phys.\  B {\bf 585} (2000) 741 [arXiv:hep-ph/0004013].
  %%CITATION = NUPHA,B585,741;%%

\bibitem{Nagy:2006xy}
  Z.~Nagy and D.~E.~Soper, Phys.\ Rev.\  D {\bf 74} (2006) 093006 [arXiv:hep-ph/0610028].
  %%CITATION = PHRVA,D74,093006;%%

\bibitem{Lazopoulos:2007ix}
  A.~Lazopoulos, K.~Melnikov and F.~Petriello, Phys.\ Rev.\  D {\bf 76} (2007) 014001 [arXiv:hep-ph/0703273].
  %%CITATION = PHRVA,D76,014001;%%

\bibitem{Anastasiou:2007qb}
  C.~Anastasiou, S.~Beerli and A.~Daleo, JHEP {\bf 0705} (2007) 071 [arXiv:hep-ph/0703282].
  %%CITATION = JHEPA,0705,071;%%

\bibitem{Bauer:2004ve}
  C.~W.~Bauer, Z.~Ligeti, M.~Luke, A.~V.~Manohar and M.~Trott,
  Phys.\ Rev.\  D {\bf 70}, (2004) 094017  [arXiv:hep-ph/0408002].
  %%CITATION = PHRVA,D70,094017;%%

\bibitem{Hoang:2005zw}
  A.~H.~Hoang and A.~V.~Manohar, Phys.\ Lett.\  B {\bf 633}, (2006) 526  [arXiv:hep-ph/0509195].
  %%CITATION = PHLTA,B633,526;%%

\bibitem{Gambino:2008fj}
  P.~Gambino and P.~Giordano, Phys.\ Lett.\ B {\bf 669} (2008) 69 [arXiv:0805.0271].
  %%CITATION = PHLTA,B669,69;%%

\bibitem{Brodsky:1982gc}
  S.~J.~Brodsky, G.~P.~Lepage and P.~B.~Mackenzie, Phys.\ Rev.\  D {\bf 28} (1983) 228.
  %%CITATION = PHRVA,D28,228;%%

\bibitem{Beneke:1994qe}
  M.~Beneke and V.~M.~Braun, Phys.\ Lett.\  B {\bf 348} (1995) 513 [arXiv:hep-ph/9411229].
  %%CITATION = PHLTA,B348,513;%%

\bibitem{Laporta:2001dd}
  S.~Laporta, Int.\ J.\ Mod.\ Phys.\ A {\bf 15} (2000) 5087 [arXiv:hep-ph/0102033].

\bibitem{Vermaseren:2000nd}
  J.~A.~M.~Vermaseren, arXiv:math-ph/0010025.

\bibitem{Ali:2007sj}
  A.~Ali, B.~D.~Pecjak and C.~Greub, Eur.\ Phys.\ J.\  C {\bf 55} (2008) 577 [arXiv:0709.4422].
  %%CITATION = EPHJA,C55,577;%%

\bibitem{Misiak:1994zw}
  M.~Misiak and M.~M{\"u}nz,
  Phys.\ Lett.\ B {\bf 344} (1995) 308 [arXiv:hep-ph/9409454].
  %%CITATION = PHLTA,B344,308;%%

\end{thebibliography}


\begin{titlepage}

\begin{flushright}
  SFB/CPP-10-39\\
  MZ-TH/10-14\\
  TTP10-26
\end{flushright}

\vskip 2cm

\centerline{\Large\bf\boldmath Complete $(O_7,O_8)$ contribution to $\bar B\to X_s\gamma$ at $O(\alpha_s^2)$}
\vskip 1cm

\begin{center}
  {\bf H.M.~Asatrian$^a$, T.~Ewerth$^{b,c}$, A.~Ferroglia$^{d,f}$, C.~Greub$^e$, and G.~Ossola$^f$}\\[1mm]
  {$^a$\sl Yerevan Physics Institute, 0036 Yerevan, Armenia}\\[1mm]
  {$^b$\sl Institut f{\"u}r Theoretische Teilchenphysik, Karlsruhe Institute of Technology (KIT),\\
    D-76128 Karlsruhe, Germany}\\[1mm]
  {$^c$\sl Dip.\ Fisica Teorica, Univ.\ di Torino \& INFN Torino, I-10125 Torino, Italy}\\[1mm]
  {$^d$\sl Institut f{\"u}r Physik (THEP), Johannes Gutenberg-Universit{\"a}t\\ D-55099 Mainz, Germany}\\[1mm]
  {$^e$\sl Albert Einstein Center for Fundamental Physics, Institute for Theoretical Physics,\\
    Univ.~of Bern, CH-3012 Bern, Switzerland}\\[1mm]
  {$^f$\sl Physics Department, New York City College of Technology, 300 Jay Street,\\
    Brooklyn NY 11201, USA}
\end{center}
\medskip

\vskip 1cm

\begin{abstract}
\noindent 
%
We calculate the set of $O(\alpha_s^2)$ corrections to the branching ratio
and to the photon energy spectrum of the decay process 
$\bar B\to X_s\gamma$ originating from
the interference of  diagrams  involving the electromagnetic dipole operator
$O_7$ with diagrams involving the  chromomagnetic dipole operator  $O_8$.
The corrections evaluated here are one of the elements needed to complete
the calculations of the $\bar B\to X_s\gamma$ branching ratio at
next-to-next-to-leading order in QCD.
We conclude that this set of corrections does not
change the central value of the Standard Model prediction
for $\text{Br}(\bar B\to X_s\gamma)$ by more than $1\%$.
%
\end{abstract}

\end{titlepage}

% ---

\section{\hspace{-3mm}Introduction}

The first estimate of the $\bar{B}\to X_s\gamma$ branching ratio within the Standard Model at the
next-to-next-to-leading order (NNLO) level was published some years ago \cite{Misiak:2006zs}:
\begin{equation}\label{estimate}
  \text{Br}(\bar B\to X_s\gamma)_{{\rm SM},\,E_\gamma > 1.6\,{\rm GeV}} = (3.15\pm 0.23)\times 10^{-4}\,.
\end{equation}
This estimate combines a number of different corrections which were calculated by several groups
\cite{Misiak:2004ew,Misiak:2010sk,Gorbahn:2004my,Gorbahn:2005sa,
  Czakon:2006ss,Blokland:2005uk,Melnikov:2005bx,Asatrian:2006ph,Asatrian:2006sm,Bieri:2003ue,Misiak:2006ab}.
The prediction given in Eq.~(\ref{estimate}) must be compared with the current world averages,
\begin{equation}\label{experiment}
  \text{Br}(\bar B\to X_s\gamma)_{{\rm exp},\,E_\gamma > 1.6\,{\rm GeV}} =
  \left\{\begin{array}{l}
  \left(3.55\pm 0.24\pm 0.09\right)\times 10^{-4}\,,\mbox{\;\,(HFAG) \cite{Barberio:2008fa}}\\[1mm]
  \left(3.50\pm 0.14\pm 0.10\right)\times 10^{-4}\,,\mbox{\cite{Artuso:2009jw}}
  \end{array}\right.
\end{equation}
which include measurements from CLEO, BaBar and Belle \cite{Chen:2001fja,Aubert:2007my,Limosani:2009qg}.
The central values of the theoretical prediction and of the HFAG average are compatible at the 1.2$\sigma$
level, while both the theoretical and experimental uncertainties are very similar in size (about 7\%). Since the
experimental uncertainty is expected to decrease
to 5\% by the end of the B-factory era (which is already indicated by the average given in the second line of
Eq.~(\ref{experiment})), it is also desirable to reduce the theoretical uncertainty accordingly.

Unfortunately, at this level of accuracy, the theoretical uncertainty is dominated by non-perturbative
contributions. As long as one restricts 
the analysis to processes mediated by the electromagnetic dipole
operator
$O_7=\alpha_{\rm em}/(4\pi)m_b\left(\bar s\sigma^{\mu\nu}P_Rb\right)F_{\mu\nu}$ alone, non-perturbative effects are
well under control \cite{Falk:1993dh,Bigi:1992ne,Bauer:1997fe,Gremm:1996df,Ewerth:2009yr}. However, as soon as
operators other than $O_7$ (such as the
chromomagnetic dipole operator
$O_8=g_s/(16\pi^2)m_b\left(\bar s\sigma^{\mu\nu}P_RT^ab\right)G_{\mu\nu}^a$) are involved,
one encounters
non-perturbative effects of $O(\alpha_s\Lambda_{\rm QCD}/m_b)$.
At present, the latter   can only be estimated
\cite{Lee:2006wn}. Hence a 5\% uncertainty related to all of the unknown non-perturbative effects has been
included in Eq.~(\ref{estimate}). A further reduction of the theoretical uncertainty below
the 5\% level
seems to be rather difficult \cite{Benzke:2010js}.
Still, given the importance of Br$(\bar B\to X_s\gamma)$ in constraining physics scenarios beyond the
Standard Model \cite{Domingo:2007dx}, it is worth to reduce the perturbative uncertainties
as much as possible. 

In particular, it would be desirable to reduce the 
uncertainty associated to the interpolation in $m_c$ which was employed to obtain Eq.~(\ref{estimate})
\cite{Misiak:2006ab}.
To get rid of the interpolation in $m_c$ in the calculation of the branching ratio is a highly challenging
task and it would represent a clear improvement
of the theoretical prediction.
Indeed, considering the work that has been done since the publication of \cite{Misiak:2006zs}, and the work that
is still in progress, an update of the estimate given in Eq.~(\ref{estimate}) will soon be  warranted.
Here we
would like to mention that the effects of charm and bottom quark masses on gluon lines are now completely known
(provided that  one neglects on-shell amplitudes that are proportional to the small Wilson coefficients of the
four-quark
operators $O_3$-$O_6$) \cite{Asatrian:2006rq,Boughezal:2007ny,Ewerth:2008nv,Pak:2008qt}. Therefore this part
could be removed from the interpolation. Also the $O(\alpha_s^2\beta_0)$-effects in the $(O_2,O_2)$, $(O_2,O_7)$ and
$(O_7,O_8)$-interference, which are known  \cite{Ligeti:1999ea}, were not considered in \cite{Misiak:2006zs,
  Misiak:2006ab}. Finally,
the complete calculation of the $(O_2,O_7)$-interference for $m_c=0$ is well underway \cite{schutzmeier}.
The latter calculation in particular will help to fix the boundary for the $m_c$ interpolation for vanishing $m_c$;
this in turn would allow one   to reduce
the 3\% uncertainty in Eq.~(\ref{estimate}) due to the interpolation. For complete up-to-date lists
of needed perturbative and non-perturbative corrections to the branching ratio we refer the reader to
the reviews \cite{Misiak:2008ss,Ferroglia:2008yg,Ewerth:2009yk,Misiak:2009nr}.

In this paper we calculate the complete  $(O_7,O_8)$-interference corrections at $O(\alpha_s^2)$ to the photon
energy spectrum
$d\Gamma(b\to X^{\rm partonic}_s\gamma)/dE_\gamma$ and to the total decay width
$\Gamma(b\to X^{\rm partonic}_s\gamma)|_{E_\gamma>E_0}$, where $E_0$ denotes the lower cut in the photon energy.
The contributions containing massless and massive quark loops were already
presented in \cite{Bieri:2003ue,Ligeti:1999ea} and \cite{Ewerth:2008nv}, respectively; the contributions
which are not yet available in the literature are the ones proportional to the color factors
$C_F^2$ and $C_F C_A$. From the technical point of view, the latter are the most complicated to evaluate and
are the main subject of the present work.

The paper is organized as follows: In Sec.~2 we present our results
for the (integrated) photon energy spectrum. In Sec.~3 we provide some details
about the calculation of the corrections proportional to $ \alpha_s^2 C_F^2$ 
and $ \alpha_s^2 C_F C_A$ by analyzing the contribution of a particular Feynman diagram. The numerical impact of
the
$(O_7,O_8)$ interference on the theoretical prediction for $\text{Br}(\bar B\to X_s\gamma)$ at NNLO is
estimated in Sec.~4. Finally, we present our conclusions in Sec.~5.

% ---

\section{\hspace{-3mm}\boldmath Results for the (integrated) photon energy spectrum}

Within the low-energy effective theory, the partonic $b\to X_s\gamma$ decay rate can be written as
\begin{equation}\label{decay_rate}
  \Gamma(b\to X_s^{\rm parton}\gamma)_{E_\gamma>E_0} = \frac{G_F^2\alpha_{\rm em}
    \overline{m}_b^2(\mu)m_b^3}{32\pi^4}\,|V_{tb}^{}V_{ts}^*|^2\,\sum_{i\le j}C_i^{\rm eff}(\mu)\,
  C_j^{\rm eff}(\mu)\int_{z_0}^1\!dz\,\frac{dG_{ij}(z,\mu)}{dz}\,,
\end{equation}
where $m_b$ and $\overline{m}_b(\mu)$ denote the pole and the running $\MS$ mass of the $b$ quark, respectively,
$C_i^{\rm eff}(\mu)$ indicates the effective Wilson coefficients at the low-energy scale, $z=2E_\gamma/m_b$ is
the rescaled photon
energy, and $z_0=2E_0/m_b$ is the rescaled energy cut in the photon energy spectrum.\footnote{In this paper we assume
  that the products $C_i^{\rm eff}(\mu)\,C_j^{\rm eff}(\mu)$ are real quantities. Therefore our formulas are not
  applicable to physics scenarios beyond the Standard Model which produce complex short distance couplings.}

As already anticipated in the introduction, we will focus on the function $dG_{78}(z,\mu)/dz$ corresponding to the
interference of the electro- and the chromomagnetic dipole operators
\begin{align}
  O_7 &= \frac{e}{16\pi^2}\,\overline{m}_b(\mu)\left(\bar s\sigma^{\mu\nu}P_Rb\right)F_{\mu\nu}\,,\\[2mm]
  O_8 &= \frac{g}{16\pi^2}\,\overline{m}_b(\mu)\left(\bar s\sigma^{\mu\nu}P_RT^ab\right)G_{\mu\nu}^a\,.
\end{align}
In NNLO approximation $G_{7 8}$ can be rewritten as follows,
\begin{equation}
\label{mastereq}
  \frac{dG_{78}(z,\mu)}{dz} = \frac{\alpha_s(\mu)}{4\pi}\,C_F\widetilde Y^{(1)}(z,\mu) +
  \left(\frac{\alpha_s(\mu)}{4\pi}\right)^2C_F\widetilde Y^{(2)}(z,\mu) + O(\alpha_s^3)\,,
\end{equation}
where
$\alpha_s(\mu)$ indicates the running coupling constant in the $\MS$ scheme
and 
\begin{align}
 \widetilde Y^{(1)}(z,\mu) &= \left[\frac{2}{9}\left(33-2\pi^2\right) +
   \frac{16}{3}\,\ln\left(\frac{\mu}{m_b}\right)\right]\,\delta(1-z)\nonumber\\[2mm]
 &\qquad
 +\frac{2}{3}\left(z^2+4\right)-\frac{8}{3}\left(1-\frac{1}{z}\right)\ln(1-z)\,.
\end{align}
%\begin{equation}
%  \widetilde Y^{(1)}(z,\mu) = \left[\frac{2}{9}\left(33-2\pi^2\right) +
%    \frac{16}{3}\,\ln\left(\frac{\mu}{m_b}\right)\right]\,\delta(1-z) +
%  \frac{2}{3}\left(z^2+4\right)-\frac{8}{3}\left(1-\frac{1}{z}\right)\ln(1-z)\,.
%\end{equation}
The function $\widetilde Y^{(2)}(z,\mu)$ can  be split further 
into a sum of contributions proportional to different
color factors:
\begin{align}\label{colorf}
  \widetilde Y^{(2)}(z,\mu) &= C_F \widetilde Y^{(2,\mbox{{\tiny CF}})}(z,\mu) +
  C_A \widetilde Y^{(2,\mbox{{\tiny CA}})}(z,\mu)\nonumber\\[2mm]
  &\qquad+ T_R N_L \widetilde Y^{(2,\mbox{{\tiny NL}})}(z,\mu) + T_R N_H \widetilde Y^{(2,\mbox{{\tiny NH}})}(z,\mu) +
    T_R N_V \widetilde Y^{(2,\mbox{{\tiny NV}})}(z,\mu)\,.
\end{align}
Here, $N_L$, $N_H$ and $N_V$ denote the number of light ($m_q=0$), heavy ($m_q=m_b$), and purely virtual ($m_q=m_c$)
quark flavors, respectively; $C_F$, $C_A$
and $T_R$ are the $\mbox{SU}(3)$ color factors with numerical values
given by 4/3, 3 and 1/2, respectively. 
The expressions for the
functions $\widetilde Y^{(2,i)}(z,\mu)$ with $i=\mbox{\small NL}$, $\mbox{\small NH}$, $\mbox{\small NV}$ can be
found in \cite{Ewerth:2008nv}.
The main result of the present work  are the  so far unknown functions $\widetilde Y^{(2,i)}(z,\mu)$ with
$i=\mbox{\small CF}$,
$\mbox{\small CA}$, which  are given by
\begin{align}\label{Yabelian}
  \widetilde Y^{(2,\mbox{{\tiny CF}})}(z,\mu) &= \left(-37.1831-\frac{64}{3}\,L_\mu -
    \frac{128}{3}\,L_\mu^2\right)\,\delta(1-z) - 11.7874\,\left[\frac{\ln(1-z)}{1-z}\right]_+\nonumber\\[3mm]
  &\qquad - 20.6279\,\left[\frac{1}{1-z}\right]_+ - 41.7874\,\ln(1-z) - 6.6667\,\ln^2(1-z)\nonumber\\[3mm]
  &\qquad + f_1(z) - 12\,\widetilde Y^{(1)}(z,m_b)L_\mu + \frac{64}{3} H^{(1)}(z,m_b)L_\mu\,,
\end{align}
%
\begin{align}\label{Ynonab}
  \widetilde Y^{(2,\mbox{{\tiny CA}})}(z,\mu) &= \left(4.7666+\frac{808}{27}\,L_\mu +
    \frac{272}{9}\,L_\mu^2\right)\,\delta(1-z)\nonumber\\[2mm]
  &\qquad - 6.5024\,\ln(1-z) + f_2(z) + \frac{34}{3}\,\widetilde Y^{(1)}(z,m_b)L_\mu\,,
\end{align}
where
%
\begin{align}
 H^{(1)}(z,m_b) &= { -\left(\frac{5}{4}+\frac{\pi^2}{3}\right)}\,
 \delta(1-z) - \left[\frac{\ln(1-z)}{1-z} \right]_+\nonumber\\[1mm]
 &\qquad -{\frac{7}{4}}\left[\frac{1}{1-z} \right]_+\!\!-
 \frac{z+1}{2}\,\ln(1-z) + \frac{7+ z - 2 z^2}{4}\,, 
\end{align}
%
\begin{align}
  f_1(z) &= 20.6279 - 108.484\,z + 13.264\,z^2 + 16.1268\,z^3 - 33.2188\,z^4\nonumber\\[1mm]
  &\qquad + 69.8819\,z^5 - 111.088\,z^6 + 118.405\,z^7 - 79.6963\,z^8 + 29.929\,z^9\nonumber\\[1mm]
  &\qquad - 4.76579\,z^{10} - 56.8265\,(1-z)\ln(1-z)\nonumber\\[1mm]
  &\qquad - 8.11265\,(1-z)\ln^2(1-z) - 5.77146\,(1-z)\ln^3(1-z)\,,
\end{align}
%
\begin{align}
  f_2(z) &= 17.0559\,z + 20.9072\,z^2 - 0.471626\,z^3 + 10.1494\,z^4\nonumber\\[1mm]
  &\qquad - 17.4241\,z^5 + 24.7733\,z^6 - 20.4582\,z^7 + 8.47394\,z^8 - 0.173599\,z^9\nonumber\\[1mm]
  &\qquad - 0.657813\,z^{10} + 5.66536\,(1-z)\ln(1-z)\nonumber\\[1mm]
  &\qquad - 11.1319\,(1-z)\ln^2(1-z) + 1.3999\,(1-z)\ln^3(1-z)\,.
\end{align}
%
Note that function $H^{(1)}(z,\mu)$ also appeared in Eq.~(2.11) of Ref.~\cite{Asatrian:2006sm},
and that we introduced the short-hand notation $L_\mu=\ln(\mu/m_b)$.

In Eqs.~(\ref{Yabelian}) and (\ref{Ynonab}) the $z$-dependence of the $\mu$-dependent terms and of those
terms which become singular in the limit $z \to 1$ is exact. The functions $f_1(z)$ and $f_2(z)$ were
instead obtained by making an ansatz 
for our numerical results of the non-singular parts, using the 
functional form
\begin{equation}
\label{fitform}
f_i(z)=\sum_{j=0}^{10} c_{i,j} \, z^j + 
c_{i,11} (1-z) \ln(1-z) +
c_{i,12} (1-z) \ln^2(1-z) +
c_{i,13} (1-z) \ln^3(1-z) \, .
\end{equation}
The coefficients $c_{0},\dots,c_{13}$ were then determined by performing a
least-square fit, using 100 specific 'data'-points. We checked that
the fit-functions remain essentially the same when changing the set of
data-points. In particular, the integrals of the fit-functions, taken
in an interval $[z_0,1]$ $(0 \le z_0 < 1)$, remain basically
unchanged. The same holds true when changing the functional ansatz given in Eq.~(\ref{fitform}), 
e.g., to contain additional terms proportional to $(1-z)^2 \, \ln^{n}(1-z)$, with $n=1,2,3$.

The plus distributions appearing in Eq.~(\ref{Yabelian}) are defined as
\begin{equation}\label{plusdis1}
  \int_0^1\!dz\left[\frac{\ln^n(1-z)}{1-z}\right]_+g(z) =
  \int_0^1\!dz\,\frac{\ln^n(1-z)}{1-z}\left[g(z)-g(1)\right]\,,
\end{equation}
where $g(z)$ is an arbitrary test function which is regular at $z=1$, and $n=0,1$. In case the
integration does not include the endpoint $z=1$, we have ($c<1$)
\begin{equation}\label{plusdis2}
  \int_0^c\!dz\left[\frac{\ln^n(1-z)}{1-z}\right]_+g(z) =
  \int_0^c\!dz\,\frac{\ln^n(1-z)}{1-z} g(z)\,.
\end{equation}

We observe that the plus distributions are present only in the part of the spectrum proportional to $C_F^2$,
see Eq.~(\ref{Yabelian}).
This is in agreement with the results reported in \cite{Neubert:2004dd};
following the procedure presented in that work, it is possible to conclude 
that the plus distributions appearing in the $(O_7,O_8)$ component of the 
photon energy spectrum at ${\mathcal O (\alpha_s^2)}$ must be the same ones as in the $(O_7,O_7)$
component of the spectrum at  ${\mathcal O (\alpha_s)}$ (up to an overall factor). In particular
\bea
\left. \widetilde Y^{(2,\mbox{{\tiny CF}})}(z,m_b)\right|_{\mbox{{\footnotesize
plus distrib.}}}  &=& - \frac{8}{9} \left( 33 - 2 \pi^2 \right) \left\{   
\left[\frac{\ln(1-z)}{1-z} \right]_+ + 
\frac{7}{4} \left[\frac{1}{1-z} \right]_+ \right\} \, , \nn \\[2mm]
&=& - 11.7874 \left[\frac{\ln(1-z)}{1-z} \right]_+ - 20.6279  \left[\frac{1}{1-z} \right]_+ \, .
\label{plusdist}
\eea
The structure in Eq.~(\ref{plusdist}) emerges in our diagrammatic
calculation from delicate cancellations 
among several contributions, and it provides a valuable 
test for our result.

% --

\section{\hspace{-3mm}Technical details about the calculation}

In order to obtain 
the results in Eqs.~(\ref{Yabelian}) and (\ref{Ynonab}) one needs to evaluate Feynman diagrams
contributing to the process $b \to s \gamma$ up to two loops,
one-loop Feynman diagrams contributing to the process $b \to s \gamma g$,
and tree-level Feynman diagrams  contributing to the processes $b \to s \gamma g g$ (plus corresponding
diagrams involving unphysical ghosts in the final state) and $b \to s \gamma s \bar{s}$.
As discussed in detail in \cite{Asatrian:2006sm}, the interferences among the various partonic diagrams with
2, 3 and 4
particles in the final state are in one-to-one correspondence with the 2-, 3- and 4-particle cuts of the 
three-loop $b$-quarks self-energy diagrams  shown in Figs.~\ref{fermionloop}, \ref{nonabelian}, \ref{double},
\ref{abelianCA},
and \ref{abelianCF2p1}, provided that
the cut goes through the photon propagator\footnote{We only display the diagrams that contribute to
  the functions $\widetilde Y^{(2,i)}(z,\mu)$ with $i=\mbox{\small CF},\,\mbox{\small CA}$.}.
The 2-particle  cuts of the self-energy graphs correspond to the interference of a tree-level and a two-loop
diagrams for the process $b \to s \gamma$, or to the interference of two one-loop diagrams for the same process.
The 3-particle cuts split a self-energy graph into a one-loop and a tree-level
diagram contributing   to the process $b \to s \gamma g$. Finally, 
the 4-particle cuts correspond to the interference of two tree-level diagrams for the process $b \to s g g$ or
for the process $b \to s \gamma s \bar{s}$. 
 
The contribution of each single cut to the photon energy spectrum can
be obtained by employing the Cutkosky rules \cite{cut1,cut2,cut3}. In order to keep the energy of the
photon fixed, it
is necessary to insert in the integrands a factor
\cite{Melnikov:2005bx} 
\be
\delta \left(E_\gamma - \frac{p_b \cdot p_\gamma}{m_b} \right) = 
2 m_b\, \delta \left((p_b - p_\gamma)^2 - (1-z) m_b^2\right) \,,
\label{energyfix}
\ee
where $p_b$ and $p_\gamma$ denote the four momenta of the $b$ quark and the photon, respectively. 
Finally, the delta function in Eq.~(\ref{energyfix}) and all of the delta functions originating from the
Cutkosky rules for the propagators crossed by a cut can be rewritten as differences of propagators as
follows \cite{Anastasiou:2002yz,Anastasiou:2003ds},
\be
\delta\left( q^2  - m^2 \right) = \frac{1}{2 \pi i} \left( 
\frac{1}{ q^2  - m^2 -i 0} - \frac{1}{q^2  - m^2 +i 0} \right) \, .
\ee
It is then possible to evaluate all of the relevant cuts of the three-loop self-energy diagrams by first
identifying a set of Master Integrals for each cut, 
and then by 
evaluating those Master Integrals by means of the tools and techniques usually employed in the
calculation of multi-loop Feynman integrals.
As usual, we work in $d=4-2\epsilon$ space-time dimensions to regularize ultraviolet, infrared and collinear
singularities.
In the rest of this section we discuss some features of this procedure
by considering one of the simplest Feynman diagrams we encountered in the course of the
calculation. More technical details concerning the 
parameterization of the phase-space integrals can be found in
\cite{Anastasiou:2003gr,Asatrian:2006ph,Asatrian:2006sm}. 

Let us consider the topology displayed in Fig.~{\ref{ACA4top}}, which corresponds to the last diagram shown in 
Fig~{\ref{abelianCA}}.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
\begin{figure}[t]
\begin{center}
\vspace*{18mm}
\begin{picture}(0,0)(0,0)
\SetScale{1.5}
%
% core diag-Shifted
  \SetWidth{1.8}
%
\Line(-75,0)(-45,0)
%\Line(15,0)(45,0)  
\Line(45,0)(75,0)
  \SetWidth{.5}
%\Line(-15,0)(15,0)  
\Line(-45,0)(45,0)
\DashCArc(-25,0)(20,-180,0){3}
\LongArrowArc(-25,0)(26,-80,-25)
%
% specific gluons
% 
\CArc(25,0)(20,-180,0)
\LongArrowArcn(25,0)(26,-25,-75)
\CArc(0,0)(25,0,180)
\LongArrowArc(0,0)(31,110,160)
%
% operators
%
%\CBoxc(44,0)(5,5){Black}{Gray}
%\CCirc(-46,0){3}{Black}{Gray}
%
% ARROWS
\LongArrow(-70,8)(-50,8)
% TEXT
\Text(-90,18)[cb]{{\large$p_b$}}
\Text(-2,-42)[cb]{{\large $k_1$}}
\Text(-42,+37)[cb]{{\large $k_2$}}
\Text(75,-42)[cb]{{\large $k_3$}}
% CUT
  \SetWidth{1.8}
  \SetColor{Blue}
\DashLine(-35,-25)(10,30){3}
  \SetColor{Red}
\DashLine(-30,-27)(-38,15){3}
%
\end{picture}
\end{center} 
\vspace*{15mm}
\caption{{\sl A three-loop self-energy topology and the corresponding 2- and 3-particle cuts. Thin lines represent
  massless propagators; the dashed line corresponds to the photon propagator; the 2- and the 3-particle cut are
  indicated by the dashed red
  and blue lines, respectively; $k_1$, $k_2$ and $k_3$ are the loop momenta.}}\label{ACA4top}
\end{figure}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
We carried out the reduction to
Master Integrals by means of  the package AIR \cite{Anastasiou:2004vj} and by means of a private code written by
one of us (in order to have a cross check).
In the diagram corresponding to the  topology in Fig.~{\ref{ACA4top}}, only a 2- and a 3-particle cut are
present (they are indicated by the dashed red and blue lines in Fig.~{\ref{ACA4top}}, respectively). 
We will first concentrate on the
3-particle cut; the 2-particle cut will be discussed at the end of this section.
Because of the delta function in Eq.~(\ref{energyfix}), one finds ($k_1=p_\gamma$)
\be
(p_b-k_1)^2 = m_b^2 (1-z) \, ;
\ee
therefore the corresponding propagator can be immediately factored out of the integrals. The reduction indicates
that the 3-particle cut of the topology in Fig.~{\ref{ACA4top}} has three Master Integrals, which are shown
in Fig.~{\ref{MIACA4}}.
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% MIs for ACA43
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{figure}[t]
\begin{center}
\vspace*{18mm}
\begin{tabular}{ccccc}
%%%%%%%%%%%%%%%%%%
\begin{picture}(0,0)(0,0)
\SetScale{.8}
%
% core diag-Shifted
  \SetWidth{1.8}
%
\Line(-75,0)(-45,0)
%\Line(15,0)(45,0)  
\Line(45,0)(75,0)
  \SetWidth{.5}
%\Line(-15,0)(15,0)  
\Line(-45,0)(45,0)
\DashCArc(-20,0)(25,-180,0){3}
%
% specific gluons
% 
\CArc(25,0)(20,-180,0)
\CArc(-20,0)(25,0,180)
%
% operators
%
%\CBoxc(44,0)(5,5){Black}{Gray}
%\CCirc(-46,0){3}{Black}{Gray}
%\CText(0,-40){Black}{White}{ACA4}
% CUT
  \SetWidth{1.8}
  \SetColor{Blue}
\DashLine(-45,-25)(5,25){3}
%
\end{picture}
&
%--
\hspace{4cm}
&
\begin{picture}(0,0)(0,0)
\SetScale{.8}
%
% core diag-Shifted
  \SetWidth{1.8}
%
\Line(-75,0)(-45,0)
%\Line(15,0)(45,0)  
\Line(45,0)(75,0)
  \SetWidth{.5}
%\Line(-15,0)(15,0)  
\Line(-45,0)(45,0)
\DashCArc(-20,0)(25,-180,0){3}
%
% specific gluons
% 
\CArc(25,0)(20,-180,0)
\CArc(0,0)(46,0,180)
%
% operators
%
%\CBoxc(44,0)(5,5){Black}{Gray}
%\CCirc(-46,0){3}{Black}{Gray}
%\CText(0,-40){Black}{White}{ACA4}
% CUT
  \SetWidth{1.8}
  \SetColor{Blue}
\DashLine(-45,-25)(25,45){3}
%
\end{picture}
&
%--
\hspace{4cm}
&
\begin{picture}(0,0)(0,0)
\SetScale{.8}
%
% core diag-Shifted
  \SetWidth{1.8}
%
\Line(-75,0)(-45,0)
%\Line(15,0)(45,0)  
\Line(45,0)(75,0)
  \SetWidth{.5}
%\Line(-15,0)(15,0)  
\Line(-45,0)(45,0)
\DashCArc(-20,0)(25,-180,0){3}
%
% specific gluons
% 
\CArc(25,0)(20,-180,0)
%\CArc(-12,0)(34,0,25)
\CArc(-12,0)(34,0,180)
%
% operators
%
%\CBoxc(44,0)(5,5){Black}{Gray}
%\CCirc(-46,0){3}{Black}{Gray}
%\CText(0,-40){Black}{White}{ACA4}
% CUT
  \SetWidth{1.8}
  \SetColor{Blue}
\DashLine(-40,-25)(15,30){3}
%
\end{picture}
\end{tabular}
\end{center}
\vspace*{8mm}
\caption{{\sl Master Integrals for 3-particle cut of the topology displayed in Fig.~\ref{ACA4top}.}}\label{MIACA4}
\end{figure}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%

Those Master Integrals have only single poles in $\epsilon$ (as long as $z\not=1$), and are sufficiently
simple to be calculated analytically by direct integration of a
suitable Feynman parameterization of the virtual loop and of the 3-particle phase-space.
By combining the output of the reduction to Master Integrals with the analytic
expressions of the latter, it is possible to find an expression for the
contribution of this particular cut to the photon energy spectrum.
It is then straightforward to check that the coefficient of the single pole in $\epsilon$, the finite
part, and the $O(\epsilon)$ term vanish in the $z \to 0$ limit, as expected from dimensional reasoning
\cite{Misiak:2008ss}. On the contrary, in the $z \to 1$ limit this particular cut shows divergences 
of the form $\ln^n(1-z)/(1-z)$ ($n=0,1$). Such divergences give origin to the plus distribution functions
which survive in the part of the photon energy spectrum which is proportional to $C_F^2$.
To see this one needs to extract a factor $(1-z)^{b\epsilon}$ (where $b$ is an arbitrary integer) out of each
Master Integral, to combine it with possible factors $1/(1-z)$ which emerge from the reduction procedure, and
then to replace the resulting expression by using
\begin{equation}
  (1-z)^{-1+b\epsilon} = \frac{1}{b\epsilon}\,\delta(1-z) +
  \sum_{n=0}^\infty\frac{(b\epsilon)^n}{n!}\left[\frac{\ln^n(1-z)}{(1-z)}\right]_+\,.
\end{equation}
The relation above explains why one needs to calculate the cofactor
of $(1-z)^{-1+b \epsilon}$ including terms of  $O(\epsilon)$:
the latter terms contribute to the  $O(\epsilon^0)$ part of the coefficient of $\delta(1-z)$.
For the definition of the plus distributions see Eqs.~(\ref{plusdis1}) and (\ref{plusdis2}).

% ---

\input{diagramsX}

% ---

Very often we had to deal with Master Integrals which we were not able to evaluate by
direct integration of their integral representation in terms of Feynman
parameters. A powerful tool to be used in these cases is the differential
equation method \cite{diffeq,Anastasiou:2002yz,Anastasiou:2003ds,Argeri}. The goal of
the method is to employ the output of the reduction procedure for a given
topology to build differential equations which are satisfied by the Master
Integrals of that topology. In our case, we consider differential equations with
respect to $z$, the only variable which appears in the Master Integrals. 
In order to generate these differential equations, it is necessary to take the
derivative of the integrand of a given Master Integral with respect to $z$. It
is interesting to observe that the only factor in the integrand which depends on
$z$ is the propagator associated to the delta function given in Eq.~(\ref{energyfix}).

A weakness of the differential equation method is the fact that there is no
general strategy which allows to fix the integration constants which are not
determined by the solution of the differential equations.   In a number of cases it was possible to fix the
missing constants by directly evaluating the Master Integral after setting $z=0$
from the start. However, we also found Master Integrals which are singular in
$z=0$; they appear in the reduction of the 3- and 4-particle cuts of the 3rd and
the last diagram given in 4th line in Fig.~\ref{abelianCA}. To fix the
integration constants in those cases we exploited the fact  that the coefficient
of each term in the $\epsilon$ expansion of each single cut contributing to the
photon energy spectrum must vanish in the $z \to 0$ limit. However, 
we had to deal with cuts for which this procedure did
not provide enough conditions to fix all of the unknown integration constants. Therefore we
calculated some of the Master Integrals for $z=1$ to reduce the number of
unknown integration constants; subsequently we fixed the remaining ones
by considering the $z \to 0$ limit of the cuts involving those integrals. Another procedure which we
employed to reduce the number of
unknown integration constants  consists in integrating some Master
Integral over $z$ from 0 to 1.  With these methods we were able to obtain analytic
expressions for all of the poles in $\epsilon$ appearing in the calculation of
the various cuts. 
For a few cuts we calculated the finite parts only numerically. Some of the
diagrams with non-trivial endpoint behavior were checked by an independent
calculation of the integrated spectrum. We would like to mention that for the
$(O_7,O_8)$-interference the endpoint singularities are only present in 3-particle
cuts of the $b$-quark self energies; 4-particle cuts are free of endpoint
singularities since the 4-particle phase space is proportional to $(1-z)$.

Now we will turn briefly to the evaluation of the 2-particle cut in
Fig.~{\ref{ACA4top}}. The reduction to Master Integrals is carried out  along
the lines of the reduction of the 3-particle cut discussed above. However, the
insertion of the delta function given in Eq.~(\ref{energyfix}) in the integrand 
is not necessary, since the
2-particle process takes place at fixed photon energy, $z=1$. We calculated  the
four Master Integrals which appear in this case by using a numerical method
based on sector decomposition \cite{Binoth:2000ps}; this technique  allows to
disentangle overlapping infrared, collinear and ultraviolet divergences. We
applied this numerical method in order to evaluate all of the Master Integrals
arising from 2-particle cuts. However, due to the presence of internal
thresholds, the integration over some of the Feynman parameters can only be
done numerically after a suitable contour deformation
\cite{Nagy:2006xy,Lazopoulos:2007ix,Anastasiou:2007qb}.

The infrared and collinear divergences appearing in the 2-particle cuts will
cancel after adding the 3- and 4-particle cuts, which also suffer from
infrared and collinear divergences. The remaining divergences are
ultraviolet. They are removed by adding counterterm diagrams with
appropriate $Z$-factor insertions (see App.~\ref{zfac}).

% ---

\section{\hspace{-3mm}\boldmath Estimating the numerical impact on Br($\bar B\to X_s\gamma$)}

In this section we investigate the numerical size of the
$(O_7,O_8)$-interference at $O(\alpha_s^2)$ at the level
of the branching ratio of the decay process
$\bar B\to X_s\gamma$. In order to do so, we adopt the notation
and conventions introduced in \cite{Misiak:2006ab}.
The $O(\alpha_s^2)$ correction to the function $K_{78}(E_0,\mu)$  in
Eqs.~(2.6) and (3.1) of \cite{Misiak:2006ab} is given by
\begin{align}\label{k782}
  K_{78}^{(2)}(E_0,\mu) &= \frac{C_F}{2}\int_{z_0}^1\!dz\,\bigg\{\widetilde Y^{(2)}(z,\mu) -
  C_F^2\frac{8\pi}{3}\alpha_\Upsilon\delta(1-z)\nonumber\\[2mm]
  &\hspace{3cm}-C_F\left[\frac{41}{2}-2\pi^2+12\,L_\mu\right]\widetilde Y^{(1)}(z,\mu)\bigg\}\,,
\end{align}
with $\alpha_\Upsilon=0.22$.
Note that Eq.~(\ref{k782}) refers to the $1S$-scheme for the
$b$-quark mass. The value used for this parameter in the numerics 
is $m_b^{1S}=4.68$\,GeV.

Combining the results of our present paper with those of \cite{Ewerth:2008nv} we are now able to
write down the
complete expression for $K_{78}^{(2)}$ containing all abelian and non-abelian contributions as well as
the effects of the masses of the $u,\,d,\,s,\,c$ and $b$ quarks running in the bubbles inserted in gluon lines.
This complete term affects the branching
ratio by an amount 
\begin{equation}\label{delBr}
  \Delta\text{Br}(\bar B\to X_s\gamma)_{E_\gamma>E_0} = \text{Br}(\bar B\to X_ce\bar\nu)_\text{exp}
  \left|\frac{V_{tb}V_{ts}^*}{V_{cb}}\right|^2\frac{6\,\alpha_{\rm em}}{\pi\,C}\,\Delta P(E_0)\,,
\end{equation}
where
\begin{equation}
  \Delta P(E_0) = 2C_7^{(0)\text{eff}}(\mu)C_8^{(0)\text{eff}}(\mu)\left(\frac{\alpha_s(\mu)}{4\pi}\right)^2
  K_{78}^{(2)}(E_0,\mu)\,,
\end{equation}
and $C$ is the so-called semileptonic phase-space factor. In order to compare with
Ref.~\cite{Misiak:2006ab}, we employ the numerical value for $C$ which was obtained from a fit of the
measured spectrum of the $\bar B\to X_cl\bar\nu$ decay in the 1S scheme\footnote{Using the
  numerical value for $C$ as obtained from a fit in the kinetic scheme raises the central value given in
  Eq.~(\ref{estimate}) by approximately 3\% \cite{Gambino:2008fj}.}
\cite{Bauer:2004ve,Hoang:2005zw}.

One might think that $\Delta\text{Br}(\bar B\to X_s\gamma)_{E_\gamma>E_0}$
in Eq.~(\ref{delBr}) simply represents the shift due to
$K_{78}^{(2)}$ of the theoretical prediction given in Eq.~(\ref{estimate}).
This is, however, not the case because an approximated
version of $K_{78}^{(2)}$ was already included in
\cite{Misiak:2006zs}: While the
$\beta_0$-part of $K_{78}$ (i.e. $K_{78}^{(2)\beta_0}$ when following the notation of
\cite{Misiak:2006ab}) was fully taken into account, the
remaining piece, $K_{78}^{(2)\rm rem}$, was calculated for large values
of $\rho=m_c^2/m_b^2$ and then interpolated (combined with
contributions not related to the $(O_7,O_8)$-interference) to the
physical value of $\rho$. To remove $K_{78}^{(2)\rm rem}$ from the
interpolation procedure and to replace it by the exact result obtained
by us, is beyond the scope of the present paper; this issue will be
correctly treated in a systematic update of Eq.~(\ref{estimate}) in the
near future.
To get nevertheless an idea of the numerical size of the $O (\alpha_s^2)$
contribution of the $(O_7,O_8)$-interference at the level of branching ratio for $\bar B
\to X_s \gamma$, we can ignore this issue and simply discuss a few numerical aspects of the quantity
$\Delta\text{Br}(\bar B\to X_s\gamma)_{E_\gamma>E_0}$, based on
$K_{78}^{(2)}$ which we calculated in this paper.

\begin{figure}[t]
  \begin{center}
    \epsfig{figure=numerics.eps,height=6cm}\\[-2mm]
    \hspace*{.6cm}{$z_0$}\\
    \hspace*{-10.2cm}\begin{rotate}{90}\hspace*{20mm}{
        $\Delta\text{Br}(\bar B\to X_s\gamma)\cdot 10^7$}\end{rotate}
    \caption{\sl $\Delta\text{Br}(\bar B\to X_s\gamma)_{E_\gamma>E_0}$ as a function of $z_0$. See text
      for more details.}\label{plot}
  \end{center}
\end{figure}

Fig.~\ref{plot} shows $\Delta\text{Br}(\bar B\to X_s\gamma)_{E_\gamma>E_0}$ as a function of $z_0$ for
$\mu=2.5\,\text{GeV}$, $\alpha_s(2.5\,\text{GeV})=0.271$, $C_7^{(0)\rm eff}(2.5\,\text{GeV})=-0.369$,
$C_8^{(0)\rm eff}(2.5\,\text{GeV})=-0.171$,
$N_L=3$ and $N_H=N_V=1$. (Note that the scale $\mu=2.5$\,GeV defines
the central value of the branching ratio given in Eq.~(\ref{estimate}).)
The remaining
numerical input parameters are taken from \cite{Misiak:2006ab}. The thick red solid line shows the
complete result, the thin red solid line  corresponds to the contribution proportional to $C_F^2$ (including the
numerical value of $C_F^2$), the green dashed line  
indicates the contribution proportional
to $C_FC_A$ (again including $C_FC_A$), and the blue dotted line is the contribution
stemming from massless and massive quark bubbles. 
The dash-dotted blue line indicates the corrections obtained by 
applying the 
large-$\beta_0$ approximation \cite{Brodsky:1982gc,Beneke:1994qe}.
The vertical line indicates the value of $z_0$ corresponding to the 
choice $E_0 = 1.6$ GeV.
It can be seen from the figure that the contributions proportional to $C_F^2$
and $C_FC_A$ cancel each other over almost the whole range of $z_0$, resulting in a contribution of size similar
to the one due to the fermionic corrections. Only in the region very close to the endpoint the
contribution proportional to $C_F^2$ dominates, due to its singular behavior for $z_0 \to 1$.
We stress, however, that the cancellations mentioned above refer to a value of
$\mu=2.5$ GeV; they do not occur anymore when going to
smaller values of $\mu$.

For a photon energy cut-off $E_0=1.6\,\text{GeV}$, as the one employed in Eq.~(\ref{estimate}), we find
the following numerical value for the quantity in Eq.~(\ref{delBr}) (using $\mu=2.5$ GeV):
%
\begin{align}
  \Delta\text{Br}(\bar B\to X_s\gamma)_{E_\gamma>1.6\,{\rm GeV}} &=
  \frac{C_F}{2}\,\bigl(52.21\,C_F-23.57\,C_A-2.66\,C_F^2-2.28\,N_L\nn\\
  &\qquad\qquad +8.73\,N_H -1.42\,N_V\bigr)\times 10^{-7} = -3.57\cdot 10^{-7}\, ,
\end{align}
%
where in the last step we inserted numerical values for the color factors and we
set $N_L=3$ and $N_H=N_V=1$. By comparing this with the central value of the
estimate given in Eq.~(\ref{estimate}), one sees that the $O(\alpha_s^2)$
corrections $K_{78}^{(2)}$ have an impact of
$-0.11\%$ at the level of the branching ratio. For the analogous effect generated by the large-$\beta_0$
approximation of $K_{78}^{(2)}$ only, $K_{78}^{(2)\beta_0}$, we find
$+0.56\%$. These results are, however, very strongly dependent on the
scale $\mu$.
For $\mu=1.25,\,2.34,\,5$\,GeV the corresponding numbers read 
$-5.15,\,-0.23,\,-0.07\%$ (full)
and $+0.10,\,+0.49,\,+1.15\%$ (large-$\beta_0$ approximation), respectively.
From these results we conclude that the large-$\beta_0$ approximation 
does not provide a good estimate of the 
full $O(\alpha_s^2)$ correction of the $(O_7,O_8)$-contribution for $\mu\in[1.25,5]$\,GeV.
As already mentioned, 
a more detailed analysis of the effect of the complete calculation of
$K_{78}^{(2)}$ on the central 
value of Eq.~(\ref{estimate}) would require 
to repeat the interpolation
procedure of \cite{Misiak:2006ab}. While this is beyond the scope of the
present work, we can conclude that the correction originating from the 
$(O_7,O_8)$-interference at $O(\alpha_s^2)$ will
not alter the central value of Eq.~(\ref{estimate}) by more than 1\%.

% ---

\section{\hspace{-3mm}Summary and conclusions}

In the present work we calculated the set of the $O(\alpha_s^2)$
corrections to the partonic decay process $b \to X_s \gamma$ which originates from the interference  of
diagrams involving the electromagnetic dipole operator $O_7$ with 
diagrams involving the chromomagnetic dipole operator $O_8$. These corrections are one of the elements
needed in order to
complete the calculation of the  branching ratio for the radiative
decay $\bar{B} \to X_s \gamma$ up to NNLO in QCD.

To carry out the calculation, we mapped the interference of diagrams
contributing to the processes $b \to s \gamma$, $b \to s \gamma g$, $b \to s
\gamma g g$ and $b \to s \gamma s \bar{s}$ onto 2-, 3- , and
4-particle cuts for the three-loop $b$-quark self-energy diagrams which
include insertions of the operators $O_7$ and $O_8$.
Subsequently, we evaluated each single cut by employing the Cutkosky rules.
From the technical point of view, the calculation was made possible by the 
use of the Laporta Algorithm \cite{Laporta:2001dd} to identify the needed Master Integrals, 
and of the differential equation method and sector decomposition
method to solve the Master Integrals. 

From the phenomenological point of view, it is  interesting to
estimate the effect of these corrections on the 
theoretical prediction
for the  
$\bar{B} \to X_s \gamma$ branching ratio. Our conclusion is that they
will not change its central value given in Eq.~(\ref{estimate}) by more than $1\%$.

At present, the largest theoretical uncertainty affecting the
prediction in Eq.~(\ref{estimate}) is of non-perturbative origin. It is expected to set a lower limit
of about $5\%$ on the total theoretical uncertainty for the prediction of the  $\bar{B} \to X_s \gamma$
branching ratio in the near future. The perturbative $O(\alpha_s^2)$ corrections of the $(O_7,O_8)$-interference
presented in this paper are a further contribution to make the perturbative uncertainty negligible with respect
to the non-perturbative one.

% ---

\section*{\normalsize Acknowledgments}
\vspace*{-2mm}

T.E.~would like to thank M.~Misiak for helpful discussions. A.F.~is grateful to J.~Vermaseren for his kind
assistance in the use of the 
algebraic manipulation program FORM \cite{Vermaseren:2000nd}, and to G.~Paz for  discussions.
The work of H.M.A.~was partially supported by ISTC A-1606 program.
T.E.~was supported by a European Community's Marie-Curie Research Training
Network under contract MRTN-CT-2006-035505 ``Tools and Precision Calculations for Physics Discoveries at Colliders''
(until October 2009) and by DFG through SFB/TR ``Computational Particle Physics'' (from October
2009 on).
C.G.~was partially supported by the Swiss National Foundation, by EC-Contract MRTN-CT-2006-035482
(FLAVIAnet) and by the Helmholz Association through funds provided to the virtual institute ``Spin and strong QCD''
(VH-VI-231); The Albert Einstein Center for Fundamental Physics is supported by the
``Innovations- und Kooperationsprojekt C-13 of the Schweizerische Universit\"atskonferenz SUK/CRUS''.
The work of G.O. was supported by the ToK Program ``Algotools'' MTKD-CD-2004-014319 (until June 2008) and in
part by NSF Grant No.~PHY-0855489.

% ---

\appendix

\section{\hspace{-3mm}Two-particle cuts} 

To obtain the contributions to the functions
$\widetilde Y^{(1)}$, $\widetilde Y^{(2,CF)}$ and
$\widetilde Y^{(2,CA)}$ (see Eqs.~(\ref{mastereq}) and (\ref{colorf})) originating from the 2-particle cuts
of the $b$-quark self energies discussed in the paper,
we first calculated the ultraviolet renormalized on-shell
matrix elements 
\[ 
\langle O_i \rangle \equiv
\langle  s \gamma |O_{i}|b \rangle\,, \qquad \mbox{($i=7,8$)} \, .
\]
For the
$(O_7,O_8)$-interference, $\langle O_7 \rangle$ and $\langle O_8 \rangle$
are needed to one-loop and two-loop accuracy, respectively. 
By considering only the  terms proportional to $C_F^2$ and $C_FC_A$ to $O(\alpha_s^2)$, which are the ones of
interest in order to obtain Eq.~(\ref{Yabelian}) and (\ref{Ynonab}), we write
\begin{equation}
\langle O_i \rangle= \langle O_{7} \rangle_{\rm{tree}}
\left[\delta_{i7}+ \frac{\alpha_s}{4\pi} \, C_F \, D_i^{(1)}
+ \left(\frac{\alpha_s}{4\pi}\right)^2 \, C_F \, \left(C_F D_{iF}^{(2)}+ C_A D_{iA}^{(2)} + \dots\right)
+ O(\alpha_s^3)\right].
\end{equation}
Note that the operator $O_7$ in $\langle O_{7} \rangle_{\rm{tree}}$
contains the $b$-quark running mass  $\overline m_b(\mu)$.
For $O_7$ the relevant results are  \cite{Blokland:2005uk,Asatrian:2006ph}
\begin{eqnarray}
  D_7^{(1)}&=&-\frac{1}{\epsilon^2}-\frac{1}{\epsilon}\left(2\,L_\mu+2.5\right) -2 \,L_\mu^2-7\,L_\mu -
  6.8225\nonumber\\[1mm]
  &&-\,\epsilon\,\left(1.3333 \,L_\mu^3 + 7 \,L_\mu^2 + 13.6449 \,L_\mu +13.4779\right)\nonumber\\[2mm]
  &&-\,\epsilon^2\,\left(0.6667 \,L_\mu^4+ 4.6667 \,L_\mu^3 + 13.6449 \,L_\mu^2 + 26.9559 \,L_\mu + 26.1412\right), 
\end{eqnarray}
and for  $O_8$ one finds
\begin{eqnarray}
  D_{8}^{(1)}&=&2.6667 \,L_\mu + 1.4734 + 2.0944\,i\nonumber\\[2mm]
  &&\quad+\,\epsilon\,\left[2.6667 \,L_\mu^2+ 2.9468 \,L_\mu-1.1947+\,i\,(4.1888\,L_\mu + 4.1888)\right]
  \nonumber\\[2mm]
  &&\quad+\,\epsilon^2\big[1.7778\,L_\mu^3+2.9468 \,L_\mu^2-2.3894 \,L_\mu -5.5373\nonumber\\[2mm]
    &&\hspace{1.8cm}+\,i\,\left(4.1888 \,L_\mu^2+8.3776 \,L_\mu +2.1627\right)\big],\nonumber\\[3mm]
  D_{8F}^{(2)}&=&
  D_8^{(1)}\left(-\frac{1}{\epsilon^2}-\frac{1}{\epsilon}\left(2\,L_\mu+2.5\right)\right)-5.3333 \,L_\mu^3
  -32.2802 \,L_\mu^2-50.9612 \,L_\mu\nonumber\\[1mm]
  &&\quad -1.8875-i\left(4.1888 \,L_\mu^2+31.4159 \,L_\mu +29.8299\right),\nonumber\\[3mm]
  D_{8A}^{(2)}&=&15.111 \,L_\mu^2+31.6617 \,L_\mu + 2.38332 + i\left(23.7365 \,L_\mu +28.0745\right)\,.
\end{eqnarray}
Taking into account the phase-space factors in $d = 4 -2 \epsilon$ dimensions, we easily obtain the contributions
to the functions $\widetilde Y^{(1)}$, $\widetilde Y^{(2,\mbox{{\tiny CF}})}$ and
$\widetilde Y^{(2,\mbox{{\tiny CA}})}$ which originate from  the 2-particle cuts only:
\begin{align}
  \widetilde Y^{(1)}_{\rm 2-cuts}(z,\mu) &= \left[ \frac{2}{9} \left(33 - 2\pi^2\right)
    + \frac{16}{3} \,L_\mu \right]\,\delta(1-z)\,,\nonumber \\[2mm]
  \widetilde Y^{(2,\mbox{{\tiny CF}})}_{\rm 2-cuts}(z,\mu) &= \bigg[
    \frac{1}{\epsilon^2}\left(-5.89368 - 10.6667 \,L_\mu\right) +
    \frac{1}{\epsilon}\left(-15.849 - 72.6954 \,L_\mu - 53.3333 \,L_\mu^2\right)\nonumber \\
    &\hspace{1cm} + 3.01591- 246.244 \,L_\mu - 335.42 \,L_\mu^2 - 135.111 \,L_\mu^3\bigg]\,\delta(1-z)\,,
  \nonumber \\[2mm]
  \widetilde Y^{(2,\mbox{{\tiny CA}})}_{\rm 2-cuts}(z,\mu) &= \left[
    4.76664 + 63.3235 \,L_\mu + 30.2222 \,L_\mu^2 \right]\,\delta(1-z)\,.
\end{align}

\noindent The results given in this appendix were already used by one of us in \cite{Ali:2007sj}.

% ---

\section{\hspace{-3mm}Renormalization constants}\label{zfac}

In this appendix, we collect the  explicit expressions of the renormalization constants needed for the
ultraviolet renormalization in our calculation.
\noindent The strong coupling constant is renormalized in the $\MS$ scheme:
\be\label{alpha_ms}
 \alpha_s^{\rm bare} =
 \mu^{2\epsilon}\left(\frac{e^\gamma}{4\pi}\right)^\epsilon
 Z_\alpha^{\MS}\,\alpha_s(\mu)\,,
\ee
with
\be
 Z_\alpha^{\MS} = 1 - \frac{1}{\epsilon}\left(\frac{11}{3}\,C_A-\frac{4}{3}\,T_RN_F\right)
 \frac{\alpha_s(\mu)}{4\pi} + O(\alpha_s^2)\,,
\ee
and $N_F=N_L+N_H+N_V$. The $b$-quark mass which appears  in the operators $O_{7,8}$, as well as  the Wilson
coefficients $C_{7,8}^{\rm eff}$ themselves, are also renormalized in the $\MS$ scheme  \cite{Misiak:1994zw}:
\begin{align}
 Z_{m_b}^{\MS} &= 1 - \frac{3\,C_F}{\epsilon}\frac{\alpha_s(\mu)}{4\pi} + O(\alpha_s^2)\,,\nonumber\\[2mm]
 Z_{77}^{\MS} &= 1 + \frac{4\,C_F}{\epsilon}\frac{\alpha_s(\mu)}{4\pi} + O(\alpha_s^2)\,,\nonumber\\[2mm]
 Z_{87}^{\MS} &= - \frac{4\,C_F}{3\,\epsilon}\frac{\alpha_s(\mu)}{4\pi} +
 \bigg\{\frac{1}{\epsilon^2}\left(\frac{34}{9}\,C_FC_A-8\,C_F^2-\frac{8}{9}\,C_FT_RN_F\right)\nonumber\\[2mm]
 &\hspace{1cm} + \frac{1}{\epsilon}\left(-\frac{101}{27}\,C_AC_F+\frac{8}{3}\,C_F^2 +
 \frac{28}{27}\,C_FT_RN_F\right)\bigg\}\left(\frac{\alpha_s(\mu)}{4\pi}\right)^2 + O(\alpha_s^3)\,,\nonumber\\[2mm]
 Z_{88}^{\MS} &= 1 + \frac{2}{\epsilon}\left(4\,C_F-C_A\right)\frac{\alpha_s(\mu)}{4\pi} + O(\alpha_s^2)\,.
\end{align}

\noindent All the remaining fields and parameters are 
renormalized in the on-shell scheme. The on-shell renormalization constant for
the $b$-quark mass is given by
\be
 Z_{m_b}^{\rm OS} =
 1-C_F\,\Gamma(\epsilon)\,e^{\gamma\epsilon}\,
 \frac{3-2\epsilon}{1-2\epsilon}
 \left(\frac{\mu}{m_b}\right)^{2\epsilon}\frac{\alpha_s(\mu)}{4\pi} +
 O(\alpha_s^2)\, .
\ee
The renormalization constants for the gluon field and the $s$- and 
$b$-quark fields are
\begin{align}
  Z_3^{\rm OS} &=
   1-\frac{4}{3}\,T_R\left(N_H+N_V\rho^{-\epsilon}\right)\Gamma(\epsilon)\,e^{\gamma\epsilon}
   \left(\frac{\mu}{m_b}\right)^{2\epsilon}
   \frac{\alpha_s(\mu)}{4\pi} + O(\alpha_s^2)\,,\nonumber\\[2mm]
  Z_{2s}^{\rm OS} &= 1 + O(\alpha_s^2)\,,\nonumber\\[2mm]
  Z_{2b}^{\rm OS} &= 1 - C_F\,\Gamma(\epsilon)\,e^{\gamma\epsilon}\,\frac{3-2\epsilon}{1-2\epsilon}
  \left(\frac{\mu}{m_b}\right)^{2\epsilon}\frac{\alpha_s(\mu)}{4\pi} + O(\alpha_s^2)\,,
\end{align}
where $\rho=m_c^2/m_b^2$.

% ---



\end{document}
